Author's response to reviews

**Title:** Clinicopathologic characteristics and prognostic factors of ovarian fibrosarcoma: the results of a multi-center retrospective study

**Authors:**

- Long Huang (huanglong914@gmail.com)
- Ling-Min Liao (253089409@qq.com)
- Hui-Yun Wang (wanghuiyun@gmail.com)
- Min Zheng (386006639@qq.com)

**Version:** 3  **Date:** 26 October 2010

**Author's response to reviews:** see over
Dear Editor,

Thank you for your acceptance of our manuscript for publication in BMC Cancer (MS: 1038793055403434). We appreciate the concerns and suggestions provided by you. We have revised our manuscript accordingly. Our point-by-point description of the changes made is provided below.

1: we have checked the email addresses.

2: we have changed author’s full name to initials in the authors’ contributions section.

3: we have pasted the tables to the manuscript following the references.

4: we compiled the figure according to the figure guide online.

Sincerely yours

Min Zheng, State Key Laboratory of Oncology in Southern China and Department of Gynecology, Cancer Center, Sun Yat-sen University, 651 Dongfeng Road East, Guangzhou, Guangdong 510060, P.R.China

Tel: 86-20-32968086, Fax: 86-20-32968086

E-mail: 386006639@qq.com

2010.10.26